Author’s response to reviews

Title: Dynamic three-dimensional echocardiography combined with semi-automated border detection offers advantages for assessment of resynchronization therapy

Authors:

Dr Tamas Szili-Torok (t.szilitorok@erasmusmc.nl)
Mr Boudewijn J Krenning (b.krenning@erasmusmc.nl)
Marco M Voormolen (m.m.voormolen@erasmusmc.nl)
Jos RTC Roelandt (j.r.t.c.roelandt@erasmusmc.nl)

Version: 2 Date: 8 Aug 2003

PDF covering letter
Dear Editor-in-Chief,

Please find the revised manuscript, which has been corrected according to the reviewer’s comments for publication by your journal. We agreed with most of the issues raised and we hope that we have addressed them satisfactorily. This has certainly improved the paper. Our detailed answers to the reviewer’s comments are presented on separate sheets.

We hereby certify that this paper is not under consideration by any other publication and has not been published elsewhere. All of the co-authors participated in the preparation, have seen the manuscript and agree with its content and the conclusions of the work. There is no conflict of interest in publishing this paper.

Yours sincerely,

T Szili-Torok
(on behalf of the authors)
Detailed answer for the comments of reviewer 1:

1. As suggested by this review we added references to the paper.

2. Potential advantages of 3D imaging are incorporated into the revised manuscript.
Detailed answers for reviewer 2.

1. The feasibility nature of the paper is more emphasized in the revised manuscript, as suggested by this reviewer.

2. Excellent point of this reviewer, whether this technique is applicable for atrial fibrillation. Since the acquisition requires stable RR intervals, in its present form it would not be accurate. However, with the use of real-time 3D transducer this technical problem may overcome.